Author's response to reviews

Title: The importance of the concepts of disaster, catastrophe, violence, trauma and barbarism in defining posttraumatic stress disorder in clinical practice

Authors:

Luciana L Braga (lorensblu@gmail.com)
José P Fiks (jpfiks@uol.com.br)
Jair J Mari (jamari@attglobal.net)
Marcelo F Mello (mf-mello@uol.com.br)

Version: 4 Date: 8 August 2008

Author's response to reviews:

1) Remotion of "Braga" from the head of each page;
2) Revision of the Affiliations of Authors;
3) Revision of References and confirmation that they are all mentioned in the current version of the text.